Author’s response to reviews

Title: Systematic review of prognostic models in traumatic brain injury

Authors:

Pablo Perel (pablo.perel@lshtm.ac.uk)
Phil Edwards (phil.edwards@lshtm.ac.uk)
Reinhard Wentz (reinhard.wentz@lshtm.ac.uk)
Ian Roberts (ian.roberts@lshtm.ac.uk)

Version: 8 Date: 13 November 2006

Author’s response to reviews:

Dear editor
We have now made all the changes asked in your recent e-mail (13-11-06)

We listed the authors in order and underneath provided affiliation details linked to the authors by superscript numbers. Qualifications and job titles were removed from the title page. Authors e-mail addresses were listed under the affiliation details.

We renamed the section Introduction as 'Background'

We included the figure number in the Figure legends. We preceded the figure title with 'Figure 1' and refered to the figure as figure 1 throughout the manuscript.

We removed unnecessary capitals for the title of Table 1

we removed the reference to table 3.3.

Many thanks

with best regards

Pablo Perel